Author's response to reviews

Title: Eosinophilic gastroenteritis in a young girl - Long term remission under Montelukast

Authors:

Ivo Quack (ivo.quack@rub.de)
Lorenz Sellin (lorenz.sellin@rub.de)
Nikolaus J Buchner (nikolaus.buechner@rub.de)
Dirk Theegarten (dirk.theegarten@rub.de)
Lars C Rump (christian.rump@rub.de)
Bernhard F Henning (bernhard.henning@rub.de)

Version: 3  Date: 14 June 2005

Author's response to reviews: see over
Dear BioMed Central Editorial Team,

please find enclosed the changes according to the reviewer’s comment. The revised manuscript has been uploaded.

**Comments to reviewer:**

**Major Revisions:**
Conclusion has been revised. First report of follow-up >20 month of Schwartz et al. has been acknowledged.

**Minor Revisions:**
Correction of spelling errors, authors affiliation changed (due to name change of our department)

**Discretionary Revisions:**
None stated

For any further questions please don’t hesitate to contact me.

On behalf of all authors.

Best regards,

Ivo Quack